Author’s response to reviews

Title: Locomotion and muscle mass measures in a murine model of collagen-induced arthritis

Authors:

Anita Hartog (Anita.Hartog@danone.com)
Judith Hulsman (Judith.Hulsman@wur.nl)
Johan Garssen (J.Garssen@uu.nl)

Version: 2 Date: 19 May 2009

Author’s response to reviews:

Dear Editor,

The manuscript has been revised conform the manuscript formatting checklist. No other changes have been made.

Kind regards,
Anita Hartog